Author's response to reviews

Title: Young women's decisions to accept chlamydia screening: influences of stigma and doctor-patient interactions.

Authors:

Myles Balfe (balfem@yahoo.com)
Ruairi Brugha (rbrugha@rcsi.ie)
Diarmuid O' Donovan (diarmuid.odonovan@nuigalway.ie)
Emer O' Connell (emer.oconnell@hse.ie)
Deirdre Vaughan (deirdre.vaughan@nuigalway.ie)

Version: 4 Date: 19 July 2010

Author's response to reviews:

Manuscript: 1280750738358352

Dear Sir/madam,

Please find attached a copy of our revised paper ‘Young women’s decisions to accept chlamydia screening: influences of stigma and doctor-patient interactions’. We have the editorial changes to the paper that you requested:

We have removed the key words from the paper.
We have removed underlining from the paper.
Blanks pages have been removed.
We have checked the typography.
We now use initials rather than fullnames in the authors' contributions sections.

Thank you
Myles